Author’s response to reviews

Title: Crude extract from Libidibia ferrea Linn leaves decreased intra articular inflammation induced by zymosan in rats

Authors:
Tamires Falção (tamiresrfalcao@hotmail.com)
Cassio Rodrigues (auriprinino@gmail.com)
Aurigena Araújo (aurigena@ufrnet.br)
Caroline Medeiros (carolineaddisonfarma@yahoo.com.br)
Luiz Soares (phtech@uol.com.br)
Magda Ferreira (magda_nani@hotmail.com)
Roseane Vasconcelos (roseane2202@gmail.com)
Raimundo Araújo Jr (araujojr@cb.ufrn.br)
Gerlane Guerra (gerlaneguerra@hotmail.com)

Version: 1 Date: 23 Apr 2018

Author’s response to reviews:

April 23 th, 2018

BCAM-D-18-00555R1

Crude extract from Libidibia ferrea Linn leaves decreased intra articular inflammation induced by zymosan in rats

Dear Editor,

We are submitting a revised version of our manuscript entitled “Crude extract from Libidibia ferrea Linn leaves decreased intra articular inflammation induced by zymosan in rats ”, according to yours and the reviewers comments and suggestions.
We would like to thank you and the reviewers for taking time in reading and suggesting modifications to our manuscript. The suggestions and feedback have been very useful to improve the quality of our article.

With best regards,

Aurigena Antunes de Araújo

Corresponding author (On behalf of the authors of the manuscript)